Author’s response to reviews

Title: SGLT2 inhibitors in T2D and associated comorbidities — differentiating within the class

Authors:

Guntram Schernthaner (guntram.schernthaner@meduniwien.ac.at)
Heinz Drexel (heinz.drexel@extern.insel.ch)
Evgeny Moshkovich (evgenym@clalit.co.il)
Birute Zilaitiene (zilaitiene@gmail.com)
Emil Martinka (martinkaemil@hotmail.com)
Leszek Czupryniak (leszek.czupryniak@wum.edu.pl)
Tamás Várkonyi (varkonyitamas@gmail.com)
Andrej Janež (andrej.janez@kclj.si)
Kristine Ducena (kducena@gmail.com)
Katarina Lalić (katarina.s.lalic@gmail.com)
Tsvetalina Tankova (tankova@iname.com)
Martin Prazny (mpra@lf1.cuni.cz)
Lea Duvnjak (lduvnjak@idb.hr)
Olga Sukhareva (olgasukhareva@mail.ru)
Harald Sourij (harald.sourij@cbmed.at)

Version: 2 Date: 23 May 2019

Author’s response to reviews:

Point-by-point rebuttal

Schernthaner et al. submitted to BMC Endocrine Disorders
Reviewer 1 (Gabriel Chodick)

All comments were adequately addressed

> We thank the reviewer for acknowledging that we have addressed all concerns raised.

Reviewer 2

> There were no comments from Reviewer 2.

Editor comments

1. Authors’ contributions

We note that two authors have the same initials (EM). To distinguish between them in the Authors’ Contributions section please designate them as EM1 and EM2, with EM1 corresponding to Evgeny Moshkovich, chosen because this name is furthest up the author list.

> Thank you for drawing our attention to this. The author contributions list has now been updated. We have also similarly updated the 'competing interests' section.

2. Clean copy

On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please remove ‘SGLT2s in T2D cover letter-REVISED MANUSCRIPT.docx’ and ‘Point by point rebuttal.docx’ from the file inventory as these are no longer required at this stage.

> These additional files have now been removed. There is no tracked changing or highlighting in the uploaded manuscript.